Author's response to reviews

Title: Comparison of a French pediatric type 1 diabetes cohort's responders and non-responders to an environmental questionnaire

Authors:

Sophie Le Fur (sophie.le-fur@inserm.fr)
Isis-Diab Network (info@isis-diab.org)
Pierre Bougnères (pierre.bouqneres@inserm.fr)
Alain-Jacques Valleron (alain-jacques.valleron@inserm.fr)

Version: 2  Date: 18 July 2014

Author's response to reviews:

Dear Editor,

According to your recommandations, we have done the following changes in our manuscript:

(1) We have included the email addresses of all the authors in the title page of the manuscript
(2) We have checked the abstract format and length (no changes, all is right)
(3) We have checked the report of ethical approval and consent in our manuscript (no changes, all is right in the Methods' section)
(4) We have added pages' and lines' numbers to facilitate reviewing.

We hope this revised version will suit you and we stay at your disposal if needed.

Sincerely yours,

Sophie Le Fur